Dear editor,
Thank you for the correspondence and for the reviews to my manuscript. I have taken positively the critique and suggestions recommended by the reviewers and this re-submission bears corrections to the same. Please find herewith my specific corrections, explanations and adjustments as itemized in the bullets below:

1. The section on authors' contribution has been added at the bottom of the document. M.F and D.C are acknowledged and are not co-authors
2. The ethnicity of the patient has been included in the new version of revised manuscript
3. Appropriate wording has been used in the statements in sections on genetic linkage, inheritance and bilaterality. Appropriate citation has been included in these cases (referee 1 recommendation)
4. Review of literature and a tabular presentation of all relevant case reports for the last ten years as searched from pubmed TM .These results have helped the author demonstrate easily the nature of reported extra ocular associations of peters anomaly and bring to the fore the unique array of associations (skeletal) in this case report.

As earlier regretted, the patient presented in this case report died and therefore the priceless idea of a second level investigation cannot be materialized. Thank you for giving me the honor of having my case peer reviewed in this highly ranked journal, I look forward to hearing from you.